



BAISHIDENG PUBLISHING GROUP INC

8226 Regency Drive, Pleasanton, CA 94588, USA
Telephone: +1-925-223-8242 Fax: +1-925-223-8243
E-mail: bpgoffice@wjgnet.com http://www.wjgnet.com

ESPS PEER-REVIEW REPORT

Name of journal: World Journal of Experimental Medicine
ESPS manuscript NO: 32731
Title: Recent Clinical Trials of Cancer Immunogene Therapy in Companion Animals
Reviewer's code: 00291404
Reviewer's country: United States
Science editor: Xiu-Xia Song
Date sent for review: 2017-01-21 21:32
Date reviewed: 2017-01-22 03:17

CLASSIFICATION	LANGUAGE EVALUATION	SCIENTIFIC MISCONDUCT	CONCLUSION
<input type="checkbox"/> Grade A: Excellent	<input type="checkbox"/> Grade A: Priority publishing	GoogleSearch:	<input type="checkbox"/> Accept
<input type="checkbox"/> Grade B: Very good	<input type="checkbox"/> Grade B: Minor language polishing	<input type="checkbox"/> The same title	<input type="checkbox"/> High priority for publication
<input type="checkbox"/> Grade C: Good	<input type="checkbox"/> Grade C: A great deal of language polishing	<input type="checkbox"/> Duplicate publication	<input type="checkbox"/> Rejection
<input type="checkbox"/> Grade D: Fair	<input type="checkbox"/> Grade D: Rejected	<input type="checkbox"/> Plagiarism	<input type="checkbox"/> Minor revision
<input type="checkbox"/> Grade E: Poor		<input type="checkbox"/> No	<input type="checkbox"/> Major revision
		BPG Search:	
		<input type="checkbox"/> The same title	
		<input type="checkbox"/> Duplicate publication	
		<input type="checkbox"/> Plagiarism	
		<input type="checkbox"/> No	

COMMENTS TO AUTHORS

The authors have summarized the progress of clinical studies of using immunogene therapy alone or in combination with chemotherapy or radiation of pets with spontaneous tumors in the last three years (2014-2016). As an investigator working on gene therapy and immunotherapy for cancers in mouse and in clinical trials for human cancer patients, this reviewer found that the mini-review is well written and valuable to many readers in the field.

ANSWERS TO REVIEWER 00291404

Thank you very much for your comments. It is glad to know that as an investigator working on gene therapy and immunotherapy for cancers you find that this work may be valuable to many readers in the field.



BAISHIDENG PUBLISHING GROUP INC

8226 Regency Drive, Pleasanton, CA 94588, USA

Telephone: +1-925-223-8242

Fax: +1-925-223-8243

E-mail: bpgoffice@wjgnet.com

http://www.wjgnet.com

ESPS PEER-REVIEW REPORT

Name of journal: World Journal of Experimental Medicine

ESPS manuscript NO: 32731

Title: Recent Clinical Trials of Cancer Immunogene Therapy in Companion Animals

Reviewer's code: 00646393

Reviewer's country: United States

Science editor: Xiu-Xia Song

Date sent for review: 2017-01-21 21:32

Date reviewed: 2017-02-14 05:57

CLASSIFICATION	LANGUAGE EVALUATION	SCIENTIFIC MISCONDUCT	CONCLUSION
<input type="checkbox"/> Grade A: Excellent	<input type="checkbox"/> Grade A: Priority publishing	GoogleSearch:	<input type="checkbox"/> Accept
<input type="checkbox"/> Grade B: Very good	<input type="checkbox"/> Grade B: Minor language polishing	<input type="checkbox"/> The same title	<input type="checkbox"/> High priority for publication
<input type="checkbox"/> Grade C: Good		<input type="checkbox"/> Duplicate publication	
<input type="checkbox"/> Grade D: Fair	<input type="checkbox"/> Grade C: A great deal of language polishing	<input type="checkbox"/> Plagiarism	<input type="checkbox"/> Rejection
<input type="checkbox"/> Grade E: Poor	<input type="checkbox"/> Grade D: Rejected	<input type="checkbox"/> No	<input type="checkbox"/> Minor revision
		BPG Search:	<input type="checkbox"/> Major revision
		<input type="checkbox"/> The same title	
		<input type="checkbox"/> Duplicate publication	
		<input type="checkbox"/> Plagiarism	
		<input type="checkbox"/> No	

COMMENTS TO AUTHORS

This paper summarized the outcomes of immunogene therapy veterinary clinical trials published from 2014 to 2016. Although interesting, due to the specificity of this type of studies, these paper may not really provide knowledge to develop or improve therapeutic strategies for human beings. It should be submit to a more specific journal.

ANSWERS TO REVIEWER 00646393

Thank you very much for your comments. It is glad to know that you describe this work as interesting. We strengthened the translational value of veterinary clinical trials by including data on subsequently derived human clinical trials. We thoroughly corrected grammatical inconsistencies.



ESPS PEER-REVIEW REPORT

Name of journal: World Journal of Experimental Medicine

ESPS manuscript NO: 32731

Title: Recent Clinical Trials of Cancer Immunogene Therapy in Companion Animals

Reviewer’s code: 02446119

Reviewer’s country:China

Science editor: Xiu-Xia Song

Date sent for review: 2017-01-21 21:32

Date reviewed:2017-02-18 17:57

CLASSIFICATION	LANGUAGE EVALUATION	SCIENTIFIC MISCONDUCT	CONCLUSION
<input type="checkbox"/> Grade A: Excellent	<input type="checkbox"/> Grade A: Priority publishing	GoogleSearch:	<input type="checkbox"/> Accept
<input checked="" type="checkbox"/> Grade B: Very good	<input checked="" type="checkbox"/> Grade B: Minor language polishing	<input type="checkbox"/> The same title	<input type="checkbox"/> High priority for publication
<input type="checkbox"/> Grade C: Good		<input type="checkbox"/> Duplicate publication	
<input type="checkbox"/> Grade D: Fair	<input type="checkbox"/> Grade C: A great deal of language polishing	<input type="checkbox"/> Plagiarism	<input type="checkbox"/> Rejection
<input type="checkbox"/> Grade E: Poor	<input type="checkbox"/> Grade D: Rejected	<input type="checkbox"/> No	<input checked="" type="checkbox"/> Minor revision
		BPG Search:	<input type="checkbox"/> Major revision
		<input type="checkbox"/> The same title	
		<input type="checkbox"/> Duplicate publication	
		<input type="checkbox"/> Plagiarism	
		<input type="checkbox"/> No	

COMMENTS TO AUTHORS

This minireview focused on immunogene therapy of veterinary clinical trials from 2014 to 2016. The organization and writing of this manuscript is good except some grammar errors need careful check. In fact, some of the therapies have been tried in human patients. I do suggest the authors compare human data where available to the results of the veterinary clinical trials.

ANSWERS TO REVIEWER 02446119

Thank you very much for your comments. It is glad to know that you consider that the organization and writing of this manuscript is good. We thoroughly corrected grammar errors. We strengthened the translational value of veterinary clinical trials by including data on subsequently derived human clinical trials.